Author's response to reviews

Title: Ectopic internal carotid artery presenting as an oropharyngeal mass

Authors:

Emmanuel P Prokopakis MD (bourolias@hotmail.com)
Constantinos A Bourolias MD (bourolias@altecnet.gr)
Argyro J Bizakis MD (abizaki@hotmail.com)
Spyros K Karampekios MD (skarampekios@yahoo.gr)
George A Velegrakis MD (bourolias@hotmail.com)
John G Bizakis MD (bourolias@altecnet.gr)

Version: 3 Date: 24 January 2008

Author's response to reviews:

we have removed the word 'Title' from the title in the manuscript.
we have removed the authors’ qualifications and job titles (e.g. PhD / MD) from the manuscript and submissions system.
we have deleted the keywords.
we have removed any highlighting/tracking from your manuscript text.
we have changed the References